Author's response to reviews

Title: Dynamics of cinacalcet use and biochemical control in hemodialysis patients: A Retrospective New-User Cohort Design

Authors:

Diane Reams (diane.reams@unc.edu)
Paul J Dluzniewski (pdluznie@amgen.com)
Thy P Do (thypdo@gmail.com)
Susan V Yue (syue@amgen.com)
Brian D Bradbury (bradbury@amgen.com)
Abhijit V Kshirsagar (abhijit_kshirsagar@med.unc.edu)
Alan Brookhart (abrookhart@unc.edu)

Version: 4 Date: 14 October 2015

Author's response to reviews: see over
Author’s response to reviews

Title: Dynamics of cinacalcet use and biochemical control in hemodialysis patients: A Retrospective New-User Cohort Design

Authors:
B. Diane Reams, PharmD, MPH
Email: diane.reams@unc.edu

Paul J. Dluzniewski, PhD
Email: pdluznie@amgen.com

Thy P. Do, PhD
Email: thypdo@gmail.com

Susan V. Yue, MD
Email: syue@amgen.com

Brian D. Bradbury, MA, DSC
Email: bradbury@amgen.com

Abhijit V. Kshirsagar, MD, MPH
Email: abhijit_kshirsagar@med.unc.edu

M. Alan Brookhart, PhD
Corresponding author
Email: abrookhart@unc.edu

Version: 4 Date: 14 October 2015

Author’s response to reviews: see over
Reviewer's report:

1. Please clarify in the Methods if the United States Renal Data System is publicly available or if permission was required to access the data and, if so, who granted the authors permission to use the data.

   Information about permission was added to lines 121-122.

2. Format Authors Contributions according to our guidelines and to use author initials rather than names. Clarify in the Authors Contributions that all authors read and approved the final version of the manuscript.

   *http://www.biomedcentral.com/bmcgastroenterol/authors/instructions/researcharticle

   Authors contributions were reformatted, lines 292-399.

3. Include details of any funding in the Acknowledgement.

   Details about funding were added to the Acknowledgements, lines 406-409.

4. Ensure that all additional files are referenced in the main manuscript.

   All additional files are now referenced in the main manuscript. Reference to Table S7 was added, line 259.